Author’s response to reviews

Title: How to estimate glomerular filtration rate in sub-Saharan Africa: design and methods of the African Research into Kidney Diseases (ARK) study

Authors:

Robert Kalyesubula (rkalyesubula@gmail.com)
June Fabian (june.fabian@mweb.co.za)
Wisdom Nakanga (wisdomnakanga@gmail.com)
Robert Newton (rob.newton@ecsg.york.ac.uk)
Billy Ssebunnya (billy.ssebunnya@mrcuganda.org)
Josephine Prynn (josephineprynn@gmail.com)
Jaya George (jaya.george@wits.ac.za)
Alisha Wade (alisha.wade@wits.ac.za)
Janet Seeley (janet.seeley@mrcuganda.org)
Dorothea Nitsche (dorothea.nitsch@lshtm.ac.uk)
Moffat Nyirenda (moffat.nyirenda@lshtm.ac.uk)
Liam Smeeth (liam.smeeth@lshtm.ac.uk)
Sarala Naicker (saraladevi.naicker@wits.ac.za)
Amelia Crampin (mia.crampin@lshtm.ac.uk)
Laurie Tomlinson (laurie.tomlinson@lshtm.ac.uk)

Version: 1 Date: 25 May 2019

Author’s response to reviews:

BNEP-D-19-00373

How to estimate glomerular filtration rate in sub-Saharan Africa: design and methods of the African Research into Kidney Diseases (ARK) study
Robert Kalyesubula, Mmed, FISN; June Fabian; Wisdom Nakanga; Robert Newton; Billy Ssebunnya; Josephine Prynn; Jaya George; Alisha N Wade; Janet Seeley; Dorothea Nitsche; Moffat Nyirenda; Liam Smeeth; Sarala Naicker; Amelia C Crampin; Laurie A Tomlinson

BMC Nephrology

Dear Dr Kalyesubula,

Thank you for submitting your manuscript, "How to estimate glomerular filtration rate in sub-Saharan Africa: design and methods of the African Research into Kidney Diseases (ARK) study" (BNEP-D-19-00373), to BMC Nephrology.

Before it can be sent out for review, please carry out the corrections, below.

****

1. Ethics approval and consent to participate

a) Approval

We have noted that in the ethics approval and consent to participate statement that you have stated that the relevant ethical approval "will" be obtained. Please can you confirm whether ethical approval has been obtained for all participating study sites in the ethics approval and consent to participate statement. Please include the full name of the ethics committees (and the institutes to which they belong to) that approved the study to be conducted at the relevant study site and include the committee’s reference number if appropriate in the "Ethics Approval and Consent to Participate" subsection of the Declarations.

The suggested details have all been included in the section of ethics Approval and consent to participate.

b) Form of informed consent from participants

In the consent for participate statement, please confirm whether the consent you will be obtaining from study participants is written or verbal. If verbal consent will be obtained, please clearly state the reason or if this was approved by the ethics committee.

This has been clarified as written informed consent.

2. STROBE checklist

In accordance with BioMed Central editorial policies (http://www.biomedcentral.com/submissions/editorial-policies#standards+of+reporting), could you please ensure your manuscript reporting adheres to STROBE guidelines (http://www.strobe-
statement.org/) for reporting observational research. This is so your methodology can be fully evaluated and utilised. Can you please include a completed STROBE checklist as an additional file when submitting your revised manuscript.

This has been filled and submitted as additional file.

3. Clean copy

At this stage, please upload your manuscript as a clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

This has been submitted.

****